Author's response to reviews

Title: The impact of bisphosphonates on the osteoblast proliferation and Collagen gene expression in vitro

Authors:

Felix Peter Koch (koch@mkg.klinik.uni-mainz.de)
Christina Merkel (christinamerkel@gmx.net)
Thomas Ziebart (ziebart@mkg.klinik.uni-mainz.de)
Sareh Said Yekta (ssaidyekta@izkf.rwth-aachen.de)
Ralf Smeets (rasmeets@ukaachen.de)

Version: 2 Date: 31 May 2010

Head & Face Medicine May 31th 2010
Priv.-Doz. Dr. Thomas Stamm
Editor-in-Chief
(Managing Editor-in-Chief)
University of Münster
Department of Orthodontics
Priv.-Doz. Dr. Dr. Ralf Smeets
Oral and Maxillofacial Surgery
Interdisciplinary Center for Clinical Research IZKF “BIOMAT.”
University Hospital Aachen
Pauwelsstr. 30
Dear PD Dr. Stamm,

thank you for reviewing our paper entitled “The impact of bisphosphonates on the osteoblast proliferation and Collagen gene expression in vitro” by Felix Peter Koch, Christina Merkel, Thomas Ziebart, Sareh Said Yekta and Ralf Smeets.

We carefully read the reviewers’ comments and recommendations and unfortunately have to agree to the criticism. In closing we would like to express our appreciation for all the value inputs which will greatly improve the quality of this paper. All changes done in the manuscript are marked red.

Kind regards

R. Smeets MD, DMD, PhD